Author's response to reviews

Title: Reproductive health service utilization and associated factors among adolescents (15-19 years old) in Gondar Town, Northwest Ethiopia

Authors:

Senafikish A Amsalu Feleke (senafikishamsalu@gmail.com)
Zelalem B Mengesha (zelalem78@gmail.com)
Digsu N Koye (digsuneg@gmail.com)

Version: 4 Date: 1 April 2013

Author's response to reviews: see over
To BMC Health Services Research

Dear Editor,

We have been requested by the reviewers and the editor to make language corrections on our manuscript entitled “Reproductive health service utilization and associated factors among adolescents (15-19 years old) in Gondar Town, Northwest Ethiopia.” Accordingly, the whole manuscript is reviewed again by language editor of the Ethiopian Journal of Health and Biomedical Sciences. We attached both the corrected manuscript and the track change document after language revision.

Best Regards!

Senafikish Amsalu Feleke

Corresponding Author